Author's response to reviews

**Title:** Sociodemographic and occupational risk factors associated with the development of different burnout types: the cross-sectional University of Zaragoza study

**Authors:**

Jesus Montero-Marin (jmontero@unizar.es)
Javier Garcia-Campayo (jgarcamp@arrakis.es)
Marta Fajó-Pascual (mfajo@unizar.es)
Jose Miguel Carrasco (jmcarrasco.iacs@aragon.es)
Santiago Gascón (sgascon@unizar.es)
Margarita Gili (mgili@uib.es)
Fermin Mayoral-Cleries (fermin.mayoral.sspa@juntadeandalucia.es)

**Version:** 6  **Date:** 23 March 2011

**Author's response to reviews:**

We have modified the structure of the abstract adapting it to the editorial norms and we have modified it in the submission page.

In the text, we have cited the Additional files 1 and 2